

Ability of the ash dieback pathogen to reproduce and to induce damage on its host are controlled by different environmental parameters

Benoit Marçais, Arnaud Giraudel, Claude Husson

▶ To cite this version:

Benoit Marçais, Arnaud Giraudel, Claude Husson. Ability of the ash dieback pathogen to reproduce and to induce damage on its host are controlled by different environmental parameters. PLoS Pathogens, 2023, 19 (4), pp.e1010558. 10.1371/journal.ppat.1010558 . hal-04231234

HAL Id: hal-04231234 https://hal.science/hal-04231234

Submitted on 18 Oct 2023

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers.

L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution 4.0 International License



GOPEN ACCESS

Citation: Marçais B, Giraudel A, Husson C (2023) Ability of the ash dieback pathogen to reproduce and to induce damage on its host are controlled by different environmental parameters. PLoS Pathog 19(4): e1010558. https://doi.org/10.1371/journal. ppat.1010558

Editor: James K. M. Brown, John Innes Centre, UNITED KINGDOM

Received: April 30, 2022

Accepted: March 14, 2023

Published: April 20, 2023

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.ppat.1010558

Copyright: © 2023 Marçais et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Data are available in DRYAD repository (doi:10.5061/dryad. kh189328x).

RESEARCH ARTICLE

Ability of the ash dieback pathogen to reproduce and to induce damage on its host are controlled by different environmental parameters

Benoit Marçais 1*, Arnaud Giraudel¹, Claude Husson²

1 Université de Lorraine, INRAE-Grand-Est, UMR1136 Interactions Arbres/Microorganismes, Nancy, France, 2 Département de la Santé des Forêts, Ministère de l'Agriculture et de la Souveraineté Alimentaire– DGAL, Paris, France

* benoit.marcais@inrae.fr

Abstract

Ash dieback, induced by an invasive ascomycete, Hymenoscyphus fraxineus, has emerged in the late 1990s as a severe disease threatening ash populations in Europe. Future prospects for ash are improved by the existence of individuals with natural genetic resistance or tolerance to the disease and by limited disease impact in many environmental conditions where ash is common. Nevertheless, it was suggested that, even in those conditions, ash trees are infected and enable pathogen transmission. We studied the influence of climate and local environment on the ability of H. fraxineus to infect, be transmitted and cause damage on its host. We showed that healthy carriers, i.e. individuals showing no dieback but carrying H. fraxineus, exist and may play a significant role in ash dieback epidemiology. The environment strongly influenced H. fraxineus with different parameters being important depending on the life cycle stage. The ability of H. fraxineus to establish on ash leaves and to reproduce on the leaf debris in the litter (rachises) mainly depended on total precipitation in July-August and was not influenced by local tree cover. By contrast, damage to the host, and in particular shoot mortality was significantly reduced by high summer temperature in July-August and by high autumn average temperature. As a consequence, in many situations, ash trees are infected and enable H. fraxineus transmission while showing limited or even no damage. We also observed a decreasing trend of severity (leaf necrosis and shoot mortality probability) with the time of disease presence in a plot that could be significant for the future of ash dieback.

Author summary

Ash dieback is an important tree disease induced by *Hymenoscyphus fraxineus*, an invasive fungus. We studied the influence of climate and local environment on the ability of *H. fraxineus* to infect, be transmitted and cause damage on its host. We showed that healthy carriers, i.e. individuals showing no dieback but carrying *H. fraxineus*, exist and may play

Funding: BM. received grant (2014-18) from Forest Health Department, French Ministry of Agriculture and Forestry https://agriculture.gouv.fr/ le-departement-de-la-sante-des-forets-role-etmissions The funder participated in the data collection, in particular for plots in Brittany and Charente. The funder had no role in study design, analysis, decision to publish, or preparation of the manuscript BM. received grant (2018-21) from H2020 HOMED. https://homed-project.eu/ The funder had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript. BM. received grant (2021-22). Overseen by the French National Research Agency (ANR) as part of the "Investissements d'Avenir" program (ANR-11-LABX-0002-01, Lab of Excellence ARBRE) https://mycor.nancy.inra.fr/ ARBRE/ The funder had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

a significant role in ash dieback epidemiology. While the ability of *H. fraxineus* to establish and to reproduce on leaves mainly depended on summer rainfall, the damage it causes to the host depends on other parameters (high summer temperature in July-August, tree cover). As a consequence, in many situations, ash trees are infected and enable *H. fraxineus* transmission while showing limited or even no damage. This is in particular the case of non-forest situations such as hedges and isolated trees.

Introduction

The existence of healthy carriers, i.e. infected individual that remain asymptomatic, is not recognized as an important feature in plant epidemiology [1]. The existence of healthy carriers has especially been reported for weak pathogens or those that can induce damage only on stressed hosts (so-called latent pathogens, [2]). This is known to be a significant feature for important stress-induced tree diseases such as Diplodia shoot blight, sooty bark disease of maples or *Botryosphaeria* cankers [3,4,5]. However, high production of inoculum usually occurs only at symptom onset. By contrast, the existence of healthy carriers is seldom reported for primary pathogens that do not need host stress to induce significant host damages. Tolerance has been reported as a mechanism used by plants to cope with infection [6]. This mechanism postulates that some individuals will limit the impact of infection on their fitness while sustaining significant pathogen load and enabling pathogen multiplication. However, the presence of individual plants that provide important levels of inoculum while exhibiting limited or even no visible symptoms is seldom reported. Disease severity is usually assumed to correlate positively with the multiplication of the pathogen within the host [7]. Nevertheless, at a different level, when comparing different species, it has been shown that poor correlation may exist between vulnerability (reduction of fitness caused by the infection) and competence (ability to sustain the epidemic by providing large amounts of inoculum). A striking example is given by Sudden Oak Death that developed in California in the last decades: while this disease, induced by Phytophthora ramorum, affects primarily Fagaceae, killing Quercus and Lithocarpus species, the main inoculum producer is a laurel, Umbellularia californica which coexists in natural forest and does not appear to suffer much fitness reduction (so-called reservoir host, [8,9,10]).

Such a discrepancy between a pathogen's ability to establish in a stand and to produce inoculum and its ability to induce damages is an important difficulty for assessing the climatic niche of a pathogen. This assessment is a critical step to evaluate the risk associated with pathogen occurrence [11]. The most widespread approach to develop species distribution models (SDM) is to relate the species known occurrence to environmental parameters, in particular climate, through statistical analyses [12]. If large discrepancies exist between a pathogen presence and its ability to cause serious damage, this will lead to important under-reporting and to poor available data. By contrast, mechanistic SDM, which relies on the assessment of environment influence on the different stages of the pathogen cycle [13] could be a suitable alternative to assess the climatic niche of a pathogen in such conditions.

Hymenoscyphus fraxineus is an invasive pathogen that has caused ash dieback throughout Europe since the late nineties and has jeopardized *Fraxinus excelsior* stands [14]. *H. fraxineus* fulfils its life cycle on ash leaves. Those are infected by ascospores during late spring and summer and remain asymptomatic until late August when necrotic leaf lesions develop [15]. *H. fraxineus* then colonises the leaf petiole and may infect young shoots, inducing dieback [14]. Most of the inoculum production occurs on leaf debris in the forest litter (so-called rachis that encompasses the petiole and the main vein of the composite leave, on which fruit bodies i.e.

apothecia develop, [14]). By contrast, very limited inoculum production occurs on bark of diseased shoots which is considered to represent a dead-end for H. fraxineus [16]. It has been suggested that, in some conditions, H. fraxineus may reproduce on ash leaves without inducing significant dieback. Some F. excelsior individuals are not affected by H. fraxineus, even under high inoculum pressure [17]. These individuals do not sustain extensive lesions when artificially inoculated at the shoot level [18]. It was then suggested that their fallen leaves could nevertheless sustain the pathogen reproduction and that the mechanism involved in containing the disease might be tolerance instead of resistance [16]. However, no experimental data is yet available to sustain this claim. Another reported tolerance mechanism would be early abscission that could prevent transfer of the pathogen from infected leaves to shoots but still allow sexual reproduction on fallen rachises [17,19,20]. Also, environmental conditions may influence differently dieback and colonisation of the rachises in the litter. Grosdidier et al. [21] showed that in open canopy conditions, the infection levels of ash rachis by *H. fraxineus* in the litter was similar compared to forest stands with a close canopy despite much lower levels of crown dieback. Those data would suggest that healthy carriers could exist in the F. excelsior population, with individuals able to be infected in specific conditions at the leaf level and thus able to produce inoculum while sustaining very limited to no dieback.

We thus examined whether ash trees could in some conditions behave as healthy carriers. More specifically, we tested if different environmental variables may favour important stages of ash dieback, namely the ability to colonise leaf rachises and reproduce on them and the ability to induce damage on ash trees, either leaf necrosis at the end of the summer or subsequently shoots mortality. We propose that ash behaving as healthy carriers could occur in locations where the environment promotes infection and reproduction of *H. fraxineus* but not symptom development.

Results

Data were available for 55 plot*year combinations for leaf necrosis frequency (Fig 1, S1 Table). Less data were available for shoot mortality (44 site*year combinations) because some plots were lost during winter (logging, 4 plots) or were not rated in spring (S1 Table). Altogether, *H. fraxineus* was detected by the qPCR assay in 86% of the 63 tested leaves with necrosis. However, the result depended on the plot and year, with lower detection frequency in plots with very little leaf necrosis (75% of tested leaves with necrosis when overall leaf necrosis frequency was lower than 5%; 84% when it was between 5 and 15%; and 95% when it was higher than 15%).

Ability of *H. fraxineus* to complete its cycle on ash saplings with different dieback severity

This section refers to the specific study done in two of the plots, Gremecey and Seichamps in 2014–15. Most of leaf symptoms that were observed in early September 2014 were partial leaf necrosis with one or several leaflets with necrosis but most leaflets still green. No shoot mortality was observed in September 2014, the mortality occurring later, during the winter. The probability of leaf necrosis was significantly higher in Gremecey than in Seichamps (likelihood Chi-square = 4.55, p = 0.03, 38.0 ± 6.5% in Gremecey and 20.6 ± 4.4% in Seichamps). The percentage of leaves with necrosis was similar on saplings showing either no or severe dieback (dieback score of 0–1 versus 3, likelihood Chi-square of 0.7084, p = 0.40, Fig 2A). This was observed in the two studied plots as the plot * dieback status interaction had no significant effect on leaf necrosis frequency (likelihood Chi-square = 1.32, p = 0.25).



Fig 1. Location of the studied ash stands (black dot) relative to the year of first ash dieback report. Data from the Département de la Santé des Forêts. Am, Fréchencourt, Be, Sarrance, Ch1-6, Champenoux (6 plots), Co, Colombier, Fe, Ance-Féas, Gr, Gremeçey, La, Landivisiau Lu, Lupé, Ol1-2, Oloron-Sainte-Marie (2 plots), Po, Pont-de-Buis, Ro, Roche-sur-Grane, Sa, Salignac-sur-Charente Sc, Scaër, Se, Seichamps). The region border shapefile can be uploaded at https://geoservices.ign.fr/telechargement.

H. fraxineus was able to produce apothecia on rachis of both the saplings with no and severe dieback at the same rate (crown status of 0–1 versus at least 3, likelihood Chi-square = 1.19, p = 0.28, Fig 2B). The saplings with no dieback experienced limited mortality of the marked shoot while the shoot mortality was high for saplings showing dieback (likelihood Chi-square = 6.99, p = 0.01, Fig 2C). In summer 2015, the crown status of the saplings had not changed, i.e. only very minor signs of *H. fraxineus* infection were observed on saplings with no dieback symptoms in 2014.

Relationship between leaf necrosis in fall and subsequent shoot mortality

Shoot mortality in spring was strongly related to the frequency of leaves with necrosis at the end of the previous summer (Fig 3A and 3B, Table 1). It is noteworthy that a small proportion of shoots that showed no leaf necrosis in September were nevertheless dead in the next spring (Fig 3A). Shoot mortality was very different on ashes with different dieback status (Table 1, Fig 3A). The odds ratio for shoot mortality of an ash with dieback versus one without was 4.1 (95% credible interval CI [2.4, 7.0]). By contrast, leaf necrosis did not significantly depend on tree dieback status (Table 1, mean leaf necrosis frequency of $12.9 \pm 3.8\%$ for ashes without dieback and $16.3 \pm 6.3\%$ for ashes with dieback). Both leaf necrosis and shoot mortality were





strongly affected by the tree random factor with a similar range of standard deviation (Table 1). This means that, even after other variables of the model are taken into account (% tree cover, dieback status), some of the rated ashes consistently showed more leaf necrosis or shoot mortality across years. This effect was of the same magnitude for leaf necrosis and for shoot mortality.

Shoot mortality was also significantly increased at higher tree cover (Table 1, Fig 3B), with an odds ratio of 1.2 (95% CI [1.0, 1.3]) per 10% tree cover. By contrast, leaf necrosis was not affected by tree cover (Table 1). The environment random factor induced a level of variability similar for leaf necrosis and shoot mortality (Table 1). As no meso-climate effect was included in this model, high environment random factors were expected.

Early defoliation was significantly associated with shoot mortality, with shoot mortality increasing with the defoliation level (Table 1). Defoliation appeared to be a consequence of infection by *H. fraxineus* and significantly increased with leaf necrosis probability (likelihood Chisq of 58.3, p < 0.0001).

Relationship between leaf necrosis or shoot mortality and climate. The preliminary analysis showed that the best fit for foliar necrosis occurred for climatic conditions in June although the difference was minor (DIC of 99.8 compared to DIC of 100.9 for 15 May to 15 June and 113.7 for 15 June to 15 July). Likewise, for shoot mortality, the best fit occurred for climatic conditions in September to end of December (DIC of 99.8 compared to DIC of 103.5 for climatic conditions in September-October).

The model fitted adequately the data, with a proper residual distribution and an acceptable prediction of observed values in the cross-validation (S1 Fig). The main parameter influencing leaf necrosis was TX78, the average of daily maximal temperature in July and August with a strong negative effect of high summer temperatures (Table 2, Fig 4A, odds ratio of 0.70, 95% CI [0.63, 0.84]). High average temperature in June increased the probability of leaf necrosis (Table 2, odds ratio of 1.33, CI [1.11, 1.52]). By contrast, rainfall either in June or in July-



Fig 3. Modelled and observed shoot mortality for ash trees of different level of dieback. (A) Site with different dieback status (with dieback, score 0–1 or without, score of 2–3). (B) Sites with different tree cover. Points represent the observed data, pooled by increasing 10% leaf necrosis frequency. For tree cover, blue dots represent sites with 9–25 tree cover while red dots represent sites with 75–100% tree cover. Points representing less than 10 shoots are not represented.

https://doi.org/10.1371/journal.ppat.1010558.g003

Variable	Leaf necrosis [95% credible interval]	Shoot mortality [95% credible interval]
Sapling dieback status (DS)	0.27 [-0.01, 0.55]	1.27 [0.82, 1.75]
Site tree cover (TC)	-0.01 [-1.14, 1.16]	1.40 [0.07, 2.23]
Defoliation at September rating (%,DEF)	-	1.53 [0.38, 2.68]
Probability of necrotic leaves (pl)	-	1.75 [0.24, 3.27]
Tree random factor (sd, Tree.l and Tree. s)	0.93 [0.82, 1.06]	0.89 [0.55, 1.24]
Site*year random factor (sd, Env.l and Env.s)	0.95 [0.65, 1.32]	1.22 [0.97, 1.54]
Deviance	4615 [4	532, 4698]

 Table 1. Relationship between leaf necrosis and shoot mortality at the shoot level.
 Parameters of the hierarchical Bayesian model.

NOTE. Parameter with a 95% credible interval that does not contain 0 are considered significant and are marked in bold. Positive values indicate an association between high levels of both the symptom and of the environmental variable.

https://doi.org/10.1371/journal.ppat.1010558.t001

August was not related to leaf necrosis probability. Lastly, we observed a trend of decreasing leaf necrosis probability with the time of ash dieback in the area (Table 2, Fig 4C, odds ratio of 0.88, CI [0.83, 0.93]).

The shoot mortality was mainly linked to the average temperature in autumn, with mortality probability decreasing with increasing autumn temperatures (September to December, <u>Table 2</u>, Fig 4D, odds ratio of 0.69, CI [0.54, 0.86]). Rainfall in autumn was not related to shoot mortality (<u>Table 2</u>). As in the shoot level model, shoot mortality increased significantly with leaf necrosis frequency (<u>Table 2</u>). As a consequence, all parameters influencing leaf necrosis probability also influenced shoot mortality (<u>Fig 4E</u>, 4F and 4G).

Table 2.	Relationship between leaf necrosis or shoot mortality and climate parameter (plo	ot level).	Parameters of
the hiera	rchical Bayesian model.		

Variable	Symptom	Parameter [95% credible interval]
Rain June (RA6, α_2)	Leaf necrosis	0.04 [-0.16, 0.25]
Temperature June (TM6, a ₃)	Leaf necrosis	0.69 [0.37, 0.99]
Rain Junly-August (RA78 (α_4)	Leaf necrosis	-0.20 [-0.52, 0.11]
Average daily maximal temperature July-August (TX78, α5)	Leaf necrosis	-1.11 [-1.50, -0.69]
N year ash dieback presence (LDP, α6)	Leaf necrosis	-0.37 [-0.58, -0.17]
Site random factor (Site.l, sd)	Leaf necrosis	0.21 [0.01, 0.49]
Rain September-December (RA912, β_2)	Shoot mortality	-0.06 [-0.43, 0.29]
Temperature September-December (TM912, β_3)	Shoot mortality	-0.83 [-1.30, -0.39]
Site tree cover (TC, β_4)	Shoot mortality	0.77 [0.07, 1.52]
Probability of necrotic leaves (pl , β_5)	Shoot mortality	4.55 [4.38, 8.12]
Site random factor (Site.s, sd)	Shoot mortality	1.21 [0.68, 2.01]
Deviance	3745 [3667, 3826]	

NOTE. Parameter with a 95% credible interval that does not contain 0 are considered significant and are marked in bold. The variables are standardized to have zero mean and one standard deviation with the exception of pl, the leaf necrosis likelihood. Positive values indicate an association between high levels of both the symptom and of the environmental variable.

https://doi.org/10.1371/journal.ppat.1010558.t002



Fig 4. Modelled and observed leaf necrosis and shoot mortality depending on climatic parameters and sites **conditions.** The curves represent the predicted response for a given parameter, all other parameters being fixed at the average value, except for the number of year of ash dieback presence that is set to 1. The orange area (grey in printed version) represent the 95% confidence interval. Dots represent oberved values.

https://doi.org/10.1371/journal.ppat.1010558.g004

Ability of *H. fraxineus* to produce apothecia on overwintered rachises in the fructification assay

The presence of *H. fraxineus* in rachises at leaf fall was tested in 2020 and 2021 in part of the sites by qPCR. It showed striking differences between years and locations (Table 3). Moreover, it was not significantly linked to either the leaf necrosis frequency measured in September, two months before (Pearson R = 0.21, p = 0.44) or the subsequent shoot mortality (Pearson R = 0.266, pe = 0.32). In Champenoux, although the frequency of leaf necrosis was low both in 2019 and 2020, the proportion of rachis in which *H. fraxineus* was detected by qPCR was very different between the two years (Table 3). Another striking feature was that the frequency of *H. fraxineus* detection by qPCR in rachises at leaf fall was almost ten-fold higher than the frequency of necrotic leaves.

The proportion of rachises producing apothecia in the fructification assay in laboratory conditions showed very high variability, with a range of 0.01 to 0.99. This strongly depended on the year with mean values of 0.26 ± 0.11 in 2016, 0.52 ± 0.17 in 2017, 0.79 ± 0.09 in 2019 and 0.37 ± 0.23 in 2020. The proportion of rachises producing apothecia was significantly correlated to the proportion of rachises in which *H. fraxineus* was detected by qPCR at leaf fall (Pearson R = 0.71, pvalue< 0.01).

The proportion of rachises producing apothecia in the fructification assay was not significantly related to the frequency of leaf necrosis in the previous September rating (Fig 5A). It was not significantly related to the tested climatic parameters during the overwintering period (Table 4). By contrast, it was significantly related to the rainfall during the previous summer (sum of rainfall in July-August, Table 4, Fig 5B). It was also not related to the % of tree cover (Table 4).

Climate suitability in France for the different stage of H. fraxineus life cycle

Maps were produced using the developed models (Tables 2 and 4) for an average site (site random factor fixed at 0) and for years 2010 to 2020. *H. fraxineus* should be able to colonise leaves, be present on rachises shed in autumn and be able to produce apothecia in spring in most of France (Fig 6A). By contrast, its ability to produce damage should be much more limited, in particular in southern France (Fig 6B–6D). However, its ability to cause damage to ash trees is strongly dependent on local site conditions: comparison of Fig 6C and Fig 6D show large difference of predicted shoot mortality depending on the % tree cover. Uncertainty in parameter estimation very significantly affected both the maps of necrotic leaves / shoot mortality probability (S2 Fig).

Area	Year	Frequency of necrotic leaves in September	Frequency of <i>H. fraxineus</i> detection on rachises in November (qPCR)
Champenoux	2019	0.09 ± 0.03	0.87 ± 0.19
-	2020	0.05 ± 0.02	0.35 ± 0.12
Pyrenean piedmont	2020	0.12 ± 0.05	0.60 ± 0.18

Table 3. Colonisation of rachises at leaf fall in November.

https://doi.org/10.1371/journal.ppat.1010558.t003



Fig 5. Ability of *H. fraxineus* to produce apothecia on the ash rachises overwintered in common garden (fructification assay). (A) Relation with leaf necrosis observed in the same plot two months before (pvalue = 0.304). (B) Relation with climate (pvalue = 0.006, Table 4).

Discussion

We showed that carrier trees without crown dieback symptoms exist and play a significant role in ash dieback epidemiology (hereafter referred to as "healthy carriers"). *H. fraxineus* is able to establish on ash leaves and to reproduce on the leaf debris in the litter (rachises) in situations where it induces very limited shoot mortality. This can be on ash trees that show no dieback while their neighbors do show some (Figs 2 and 3A), in specific local environments (low tree cover, Fig 3B) or meso-climates (Fig 6). The climate strongly influences *H. fraxineus* with different parameters being important depending on the life cycle stage (Figs 4 and 5). As a consequence, while the pathogen might be able to establish over most France, it should to induce far less severe dieback in large parts of southern France (Fig 6). This is likely to occur in others areas of southern Europe. Last, we observed a decreasing trend of severity (leaf necrosis and shoot mortality frequency) with the time of disease presence (Fig 4D and 4H).

We were able to predict the severity of dieback in spring by observing leaf symptoms in the previous autumn. Leaf necrosis frequency in September was a good predictor of subsequent shoot mortality (Fig 3). Despite that, leaf necrosis was a very poor predictor of rachises infection at leaf fall and of the ability of *H. fraxineus* to produce apothecia, and therefore inoculum,

assay.						
Parameter	Pvalue					
-0.44	0.297					
1.23	< 0.001					
-0.29	0.353					
0.21	0.478					
0.55	0.089					
0.11	0.549					
0.0003	-					
	Parameter -0.44 1.23 -0.29 0.21 0.55 0.11 0.0003					

Table 4. Influence of climatic parameters on the ability of *H. fraxineus* to produce apothecia in the fructification assay.

Note: RA10-3 ow and TM10-3 ow, sum of rain and mean of daily temperatures at the overwintering sites from October to March.

https://doi.org/10.1371/journal.ppat.1010558.t004



Fig 6. Map of climate suitability for *H. fraxineus* ability to produce apothecia on the ash rachises or symptoms on leaf and shoot of ash trees. Safran meteorological data, 2010–20. (A). Proportion of ash rachises collected in fall producing apothecia in laboratory conditions after overwintering in a common garden (fructification assay). (B) Leaf necrosis probability. (C) Shoot mortality probability in open canopy (20% tree cover). (D) Shoot mortality probability in forest conditions (100% tree cover).

on the rachises in spring (Figs 5 and 7). This may be because of the two-month delay between leaf necrosis measurement and rachises colonisation assessment. It has to be pointed out that very little inoculum production is observed in September and October in French conditions [22]. Nevertheless, lesions present in the leaflets might have extended to the rachises. Another likely explanation is that *H. fraxineus* has the ability to asymptomatically infect ash leaves in early summer and to behave as a latent pathogen for an extended period [15]. These authors showed that leaf necrosis occurs only when *H. fraxineus* inoculum reaches a certain level in the leaf which happens at the same time as leaf senescence and of a change in the leaf necrosis also strongly promote shoot infection from the petiole (Fig 3). By contrast, high *H. fraxineus* neus levels may not be required for rachis colonisation at leaf fall in senescent tissue. This would explain why high rachis colonisation may be observed in situations with limited previous leaf necrosis (Table 3, Fig 5A).

Because high rachis colonisation may occur while symptom frequency, leaf necrosis or shoot mortality, is low (Table 3), *H. fraxineus* is able to complete its biological cycle in conditions where no significant dieback is observed and some ash trees behave as healthy carriers



Fig 7. Hypothesised impact of studied environmental parameters on *H. fraxineus* **life cycle.** Higher summer rainfall and higher spring temperatures are associated with higher rates of leaf necrosis leading to higher shot necrosis especially in high tree cover. Higher summer temperatures are associated with low leaf necrosis frequency at the end of the summer, but not to low colonisation of rachises at leaf-fall during autumn. Low autumn temperatures are associated with high shoot mortality (better transfer from leaves to shoot or lesion development in shoots?). The part of the cycle needed for inoculum production is indicated in blue while red arrows and text refers to symptoms development that does not lead to inoculum production. Cycle adapted from [41].

(Fig 7). This is the case at local level, on specific trees that remain without shoot mortality despite experiencing leaf necrosis and enabling the pathogen to complete its cycle on infected rachises (Fig 2). We did not find a difference in leaf necrosis frequency between saplings showing dieback or not (Table 1). The measure of leaf infection might not have been optimal as we only rated the presence of necrosis on the leaf without any measure of necrosis severity such as number of affected leaflets. Also, we examined only branches within reach to assess the disease severity, which may not be representative of disease severity higher in the crown. Leaf infection might be higher close to soil level as spore load in the air was shown to decrease from 1 to 3 m above soil level [23] and because microclimate is more humid close to the soil and thus more favourable to *H. fraxineus* infection.

Our experimental design did not allow us to determine whether certain ash trees showed no dieback for genetic reasons or because they were in microsites unfavorable to the disease or harbored foliar microbial competitors preventing *H. fraxineus* development. Resistance to *H. fraxineus* has been widely documented in *F. excelsior* [17,24] and thus remains a very likely explanation for at least part of the observed pattern, although different micro-climatic conditions can be another one (Fig 3A and 3B). It does not appear that some of the studied trees remained without dieback because they avoided inoculum as they showed a frequency of leaf necrosis similar to trees that showed dieback symptoms in the same plots (Figs 2 and 3). In the two stands studied in more detail in 2014, the ash density was high and the saplings remaining without dieback were in close proximity to dying saplings, questioning the possibility that they were experiencing different environmental conditions.

Nevertheless, environmental conditions strongly influenced the ability of *H. fraxineus* to infect the shoot from the leaf petiole. Indeed, although the frequency of leaf necrosis did not depend on the tree cover, the subsequent frequency of shoot mortality was much lower in hedges with a low tree cover compared to forest plots with a high tree cover (Table 1, Fig 3B). We also showed that the colonisation of rachises at leaf fall and subsequent ability of *H. fraxineus* to complete its cycle on the colonised rachises does not depend on the tree cover

(Table 3), being similar in forests conditions and in hedges in open landscape. Hence, trees with no dieback symptoms in infected stands behave as heathy carriers, allowing H. fraxineus to reproduce on their leaves as efficiently as on the leaves of neighboring trees that experience dieback. They should even be much better inoculum producers than trees affected by dieback as they retain full foliage and thus produce abundant infected rachises. This result is in agreement with the observation of Grosdidier et al. [21] that the density of infected rachises was similar in the litter beneath ash trees in hedges and in forest settings although the trees showed far less dieback symptoms in the latter situation. A possible explanation is that as the tree cover does not affect inoculum production, it does not affect infection frequency. However, at low tree cover, crowns are exposed more frequently to micro-climatic conditions adverse to H. *fraxineus* (hotter and dryer conditions, [21]), preventing the increase of the pathogen present as endophyte in leaves and thus later shoot mortality. This discrepancy between the ability to reproduce on the host and the ability to induce significant damages has been reported for other tree pathogens. For example, Diplodia sapinea mainly produce inoculum on pine cones and is widely present on cones in healthy pine stands; it however needs weakening of the trees to induce significant damages [25,26].

At a larger scale, we showed that *H. fraxineus* may be able to fulfil its cycle in regions were climate does not allow significant dieback (Fig 6). The disease has been progressing very regularly in France from 2009 to 2016 with a very noticeable slowdown of the spread in southern France from 2017 to 2019 (Fig 1). The slowdown of the spread occurred in regions with climate unfavorable to ash dieback. In 2020, the Pyrenean area was colonised while no sign of the disease had been reported north in the Garonne valley (data from the Département de la Santé des Forêts). It is possible that the pathogen was introduced via infected planted ashes [27]. However, our results show that the spread without dieback symptoms of *H. fraxineus* in the Garonne valley is also a likely hypothesis. This may have been enhanced by the very hot summers of 2018–20, which were even less favorable to *H. fraxineus*-induced shoot mortality. The delay of 3–4 years observed until the first report of ash dieback in the Pyrenean piedmont is in line with the spread observed in France of about 60 km per year [28].

The decreasing trend of leaf necrosis and shoot mortality frequency with years of disease presence may be very significant but requires confirmation. It may well be a spurious result caused by the very dry and hot climate experienced in northern and eastern France in 2018–20, at the end of the study period. Climatic conditions were taken into account in our model and high summer temperature was one of the most important feature to explain symptom severity. But the parameter we used (average daily maximal temperature) may not have accounted for the entire effect of heat waves observed during this 2018–20 period. Another possible explanation is the progressive selection of more tolerant or resistant ash trees in sites affected by ash dieback for a long time. Indeed, in the site followed for the longest period, fewer than 50% of the saplings selected in 2015 were still alive in 2021. We tried to replace dead saplings by symptomatic saplings to avoid a progressive increase in tolerant/resistant ash frequency in our sample, but this was not always possible.

Early leaf senescence was hypothesized to be a mechanism of tolerance to *H. fraxineus*, the pathogen then being shed in the leaves before getting the chance to infect shoots [17,19]. We were not able to confirm the hypothesis as early defoliation was associated with higher shoot mortality in our data. However, care has to be taken when interpreting this result. Mc Kinney et al. [17] reported that some ash trees show early leaf senescence as a genetically controlled trait, and thus show early defoliation, limiting the transfer of *H. fraxineus* from leaves to shoots. In our experiment, early defoliation was a result of infection, being tightly associated with high leaf necrosis frequency. In these conditions, transfer of the pathogen from leaves to shoots may not be hampered.

Climate appears to be a major driver of symptom induction by *H. fraxineus*. High summer temperatures was expected to limit disease severity, as the pathogen has been shown to have low survival at temperatures above 35°C [29]. Moreover, the hot summers of the Rhône valley south of Lyon limit ash dieback severity in South-East France [30]. More surprising is the low correlation of spring and summer rainfall with disease severity as it has been associated with high disease impact through increased inoculum production [31,32,33]. Spring and summer rainfall may promote infection of the leaves while H. fraxineus remains as an asymptomatic endophyte especially if high summer temperatures prevent the heavy colonisation of ash leaves by the pathogen. This could explain why spring and summer precipitations were not well correlated with frequency of symptoms such as leaf necrosis and shoot mortality (Table 2). By contrast, summer rainfall was strongly associated with both high rachis infection and ability to produce apothecia in the following spring. For example, in 2019, very limited leaf necrosis occurred in the Champenoux area because of high summer temperatures although summer rainfall allowed good colonisation by *H. fraxineus* of the rachises shed during the autumn (Table 3). This might be the likely mechanism by which spring and summer rainfall favour high ash dieback impact. The period in spring with the best fit between climatic conditions and leaf necrosis frequency was in June, although the difference with other tested periods was low. This may appear surprising as the period of ascospores production has often been shown to be in summer, in July to August [23,32,34,35]. However, many of those results correspond to central or northern Europe; in North-East France, the period of ascospores production was earlier, from mid-June to mid-July [22] and ascomata have been observed as early as the first week of May in the Pyrenean sites during this work. Shoot mortality was associated with previous leaf necrosis, but also with subsequent mild autumn temperature. This is the first report of an association of shoot mortality with autumn climatic conditions and the mechanism involved remains to be elucidated.

H. fraxineus ability to infect ash leaves and reproduce on the rachises in the forest litter does not appear to limit the pathogen ability to establish throughout France. By contrast, the low ability to induce shoot mortality explains the absence of reports of the disease in many areas of southern France. Hence, the maps showing the shoot mortality probability match well with the ash dieback risk predicted by Goberville et al. [36]. The statistical niche model produced by these authors was based on the distribution in Europe of this invasive pathogen still spreading and with a presence still limited to North-Easter France; it nevertheless well captured the potential climatic niche in France. This would confirm that south-eastern France is not favorable to ash dieback, as was suggested by Grosdidier et al. [30].

Material and methods

Sampled plots

We selected 20 plots on a temperature gradient in France (Fig 1, S1 Table) in order to have a range of summer temperature as large as possible within the range of *H. fraxineus* presence. The plots were usually studied several years in a row starting from 2013 in the area where ash dieback was first observed in 2009–10 (NE France) to 2020 in an area just recently affected by ash dieback (SW and NW France, in Brittany and Pyrenean Piedmont). The studied plots were either in forest settings or located in hedges. The ash species present was *Fraxinus excelsior* in all plots. A summary of observations of leaf necrosis and shoot mortality on the plots is given in S1 Table. Plots were characterized by their tree cover from aerial photographs (Orthophoto with 50cm pixels from IGN, http://professionnels.ign.fr). The tree cover polygons within a radius of 100 m from each plot, including all tree species, were extracted and the proportion of tree cover within that range (TC) was computed.

Survey procedure

In all studied plots, 20 ash trees were marked. In each plot, we selected suitable saplings that were contiguous, marking all those with two shoots available for rating. We usually selected saplings 2-4 m high, although occasionally, larger trees with low-lying branches were selected. Preliminary analysis showed that the tree height did not influence leaf necrosis and shoot mortality probability. Saplings of different health conditions were usually present in the plots, with a proportion of saplings without dieback of shoots varying from 20 to 75% mostly depending on the length of ash dieback presence in the area. This proportion was adequate for the hypothesis we wanted to test. Dieback only refers to the mortality of perennial organs, here the shoots; symptoms occurring on the leaves are not included. The saplings were rated annually for their crown status according to five dieback scores: 0 for complete absence of dead twigs, 1 for less than 10% of dead twigs, 2 for 10 to 50% of dead twigs, 3 for more than 50% of dead twigs and 4 for dead saplings [21]. Two annual shoots per individual were marked for rating of leaf and shoot infection. We selected the terminal shoot of the 2 lower branches that could be reached by an observer from the ground. As most saplings were 2-4 m high, this should not introduce any bias. Each sapling was rated for several years in row (1 to 6 years depending on the site and on the year of first ash dieback report, <u>S1 Table</u>). Dead twigs were replaced by the nearest shoot available from the lower part of the sapling. Dead saplings were replaced by the closest appropriate sapling, taking care to select saplings with ash dieback symptoms to avoid ending up with only individuals potentially tolerant or resistant to H. fraxineus. On average, about 18% of the saplings were replaced over the study period, mostly saplings showing dieback as they suffered high mortality.

The necrosis induced by *H. fraxineus* on leaves was rated in September. The total number of leaves present on each marked annual shoot, the number of leaves with lesions (not taking into account the number of affected leaflets) and the number of leaf scars indicating defoliation were determined (altogether 4 to 27 leaves depending on the shoot). Lesions induced by *H. fraxineus* can be recognized as brown lesions starting at leaf margin and extending preferentially on leaf veins. Leaves presenting brown lesions only on the rachis were also rated as infected. Two leaves rated as presenting *H. fraxineus*-induced lesions were sampled for further analysis in the laboratory in 2018–20. Sampled leaves were stored individually in paper bags and kept at ambient temperature. In the laboratory, they were surface sterilized (5 min in 4% NaOCl and 1 min in 90% ethanol) and analysed by qPCR for the presence of *H. fraxineus* as described by Ioos et al. [37]. In the following year's vegetation season, the status of marked shoots was assessed as living or dead. Whenever the shoot had wilted because of girdling occurring below the annual shoot, we considered that it was not possible to determine whether infection had transferred from the annual shoot leaves to the stem and the data was counted as missing.

In October-November, 20 rachises that had just fallen to the ground were sampled by site in order to assess the ability of *H. fraxineus* to produced apothecia on them (called hereafter fructification assay). The rachises were sampled at the base of the marked saplings (one rachis per sapling). This was done in all sites of Champenoux (2016, 2017, 2019, 2020), Fréchencourt (2016, 2017), Lupé (2016), Roche-sur-Grane (2017) and in the 4 Pyrenean sites (Ol1, Ol2, Fe, Be, 2020). The rachises were placed in mesh bags and overwintered in a common garden in a forest site in Champenoux (6.34251, 48.75297, EPSG 4326) except for the rachises of the plots close to Pau that were overwintered close to the site of Sarrance (-0.59692, 43.00044, EPSG 4326). This was done to minimize travel; climatic conditions during the vegetation season conditions determine inoculum production and leaf infection and we supposed that it would be the main factors controlling colonisation of the rachises. In the following spring, in April or May depending on the year, the rachises were retrieved from the mesh bags and were placed for 6–8 weeks at 18–20°C in humid chambers in large plastics boxes in contact with moistened filter paper to assess their ability to produce apothecia. The moistened filter paper was kept wet throughout the incubation and was changed weekly. Whenever a rachis produced *H. fraxineus* apothecia, the rachis part covered with pseudosclerotial plate was removed and stored in paper bags for further analysis. The part without pseudosclerotial plate was left in the humid chamber. *H. fraxineus* apothecia can be recognized as 2–4 mm white apothecia with a dark base of the pedicels and smooth margin of the fruiting disk [38]. The monitoring was stopped whenever no new production of apothecia was observed in a week (after 6–8 weeks incubation). Remaining rachises were then placed in separate bags. The amount of rachises parts producing / not producing apothecia was determined by measuring the rachis length in 2016 and 2017 and by measuring their weight after drying 3d at 50°C in 2019 and 2020. The proportion of rachises producing apothecia was then computed.

In 2019 and 2020 in sites in Champenoux and in 2020 in Pyrenean sites, 10 additional rachises were collected per site at the same time at leaf fall to determine the frequency of *H*. *fraxineus* by qPCR. Individual rachises were kept separate. They were dried 3d at 50°C, cut in small 1–2 mm sections and grinded with a mortar. A subsample of 50 mg per rachis was then used for DNA extraction with DNAeasy Plant Mini kit (Qiagen). AP1 lysis buffer (800 μ l), 4 μ l RNase, two 3-mm and twenty 2-mm glass beads were added to the tube and the sample was grounded twice using a beadbeater (Tissue Lyser, Qiagen) at 30 Hz for 30 s. The DNA extraction was then conducted according to the manufacturer's recommendations. Total DNA was eluted in 150 μ l AE buffer. The samples were then analysed by qPCR for the presence of *H*. *fraxineus* as described in Ioos et al. [37].

Ability of *H. fraxineus* to complete its cycle on ash saplings with different dieback severity

The ability of *H. fraxineus* to complete its cycle on ash saplings with no dieback symptoms was studied in two of the plots, Gremecey and Seichamps. Ash saplings about 2–3 m high were selected during summer 2014 in order to have 10–13 saplings of two types in each stand: i. saplings with either no or only very few scattered small dead shoots (no dieback, crown status of 0 or 1), ii. saplings strongly affected, with top shoots killed by *H. fraxineus* (crown status of at least 3). The two stands were densely stocked naturally regenerated ash stands heavily and uniformly infected by *H. fraxineus* for at least 4 years. The saplings without dieback were in close vicinity to individuals with severe dieback, with usually several severely affected saplings within less than 1 m from the healthy-looking sapling.

On each sapling, an annual shoot with 7–28 leaves was selected and marked at the end of July. The selected shoot infection by *H. fraxineus* was rated at the beginning of September 2014 with a record of the number of leaves with necrosis and of the total number of leaves. To measure the ability of *H. fraxineus* to complete its cycle on rachises, we placed a net around the entire crown of each sapling in mid-September to collect all the leaves falling during the autumn. The nets containing the leaves were retrieved at the end of October and placed on the ground outdoors in a nursery close to both plots to overwinter. Unfortunately, logging occurred in the Seichamps plot during autumn and a heavy machinery destroyed many studied saplings. Only four saplings without dieback and one with severe dieback were available in that plot for the net harvest. In April 2015, the rachises were retrieved from the 27 nets (11+4 without dieback and 11+1 severely affected by *H. fraxineus*) and were placed in large plastic boxes in contact with moistened filter paper. The boxes were then closed and placed in the laboratory at room temperature. After a 5-week incubation, the rachises were examined for

presence of *H. fraxineus* apothecia. The total length of each rachis was determined as well as the length between the two most distant apothecia of the rachis (hereafter called L.apo). As the part of rachises colonised by *H. fraxineus* was usually covered with many primordia of apothecium in the humid chamber, this is equivalent to the length of rachis colonised by *H. fraxineus*. The percentage of rachis colonisation on a sapling was determined as Σ *L.apo* / Σ *total rachis length*. The saplings were again assessed in summer 2015 to determine the global infection status and the presence of shoot infection on the marked shoot assessed in 2014 for leaf infection.

The likelihood of leaf necrosis in September 2014 and shoot infection in summer were assessed by logistic regression with the plot, the dieback status of the saplings and their interaction as independent variables. We used a quasibinomial distribution to analyse leaf necrosis likelihood because data showed a dispersion higher than expected for a binomial distribution (over dispersion). The percent rachis colonisation by *H. fraxineus* in spring 2015 was analysed by beta regression using the betareg package. The stand*dieback status interaction was assessed for leaf necrosis but not for shoot infection in 2015 and ability of rachises to produce apothecia as very few saplings then remained in Seichamps.

Data analysis

During survey, we noticed that some of the rated shoots were very vigorous with many rated leaves and that they were seldom infected by *H. fraxineus* whatever the general situation in the stand. These very vigorous shoots were usually sprouts from the collar of severely affected saplings. Preliminary analysis showed that shoots with more than 20 leaves had on average 5.9% necrotic leaves and no shoot mortality, compared to average values of 16.0% of necrotic leaves and 21.6% shoot mortality for the total dataset. Shoots with more than 20 leaves were thus removed from the data set prior to analysis (represent only 1.5% of the data). Two different models were developed. Bayesian models were fitted with Jags 4.3.0 emulated with R4.1.0 (R2jags library).

Relationship between leaf necrosis and shoot mortality

The first model aimed at exploring the relationship between leaf necrosis and shoot mortality at the individual shoot level and how this may be controlled by tree health and site tree cover. Data were thus kept at the shoot level for this model. A hierarchical Bayesian model was developed to assess symptoms on shoot n of sapling i in year t:

```
logit(pl[n,i,t]) = \alpha_1 + \alpha_2 \times DS[i] + \alpha_3 \times TC[i] + Tree.l[i] + Env.l[i,t]
```

```
logit(ps[n,i,t]) = \beta_1 + \beta_2 x DS[i] + \beta_3 x TC[i] + \beta_4 x DEF[n,i,t] + \beta_5 x pl[n,i,t] + Tree.s[i] + Env.s [i,t]
```

with pl[n,i,t] and ps[n,i,t] the respective probability of leaf necrosis and shoot mortality, DS[i] the ash dieback status (either no dieback, when crown rating was of 0 or 1 or with dieback, when crown status was of 2 or 3), TC[i] the percent tree cover within a radius of 100 m, DEF [n,i,t] the proportion of missing leaves at the autumn rating, Tree.l [i] and Tree.s [i] individual tree random factors and Env.l [i,t] and Env.s [i,t], plot[i]*year[t] random factors. The Env.l [i] and Env.s [i] random factors were taken as proxies for the environment, combining meso/ micro climate and plot conditions (host density, Infection history). The ash dieback status DS [i] was computed as the worst status observed on the sapling during the period of observation (from 2 to 7 years depending on the sapling). The defoliation level (DEF[n,i,t]) was assumed to influence the shoot mortality probability because Mc Kinney et al. [17] suggested that early defoliation may prevent the transfer of *H. fraxineus* from leaves to shoots and thus represent a tolerance mechanism.

Both pl[n,i,t] and ps[n,i,t] were assumed to follow binomial distribution. Non-informative priors were assigned to the model parameters according to a normal distribution N(0, 1e+05) or a uniform distribution U(0,100) for the standard deviation of the random factors Tree and Env. The model was fitted with a burn-in of 3000, for 50000 iterations with a thinning of 5. Three parallel chains with different initial parameter values were run and convergence was checked by a Gelman-Rubin test (R_{hat} were close to 1). The model was checked by looking at the plot of deviance residuals versus predicted values.

Relationship between leaf necrosis or shoot mortality and climate

The second developed hierarchical Bayesian model aimed at exploring the relationship between leaf necrosis or shoot mortality with climate. For that model, data were pooled at the plot * year level. Like in the previous model, shoot mortality was assumed to derive from former leaf infection and the proxy for leaf infection was taken as the leaf necrosis probability. Length of ash dieback presence in the area was included in the model to explain the probability of leaf necrosis. The year of first report of ash dieback by the health survey system is available by 16 x 16 km quadrat over France from data of the health survey system (Département de la Santé des Forêts, Fig 1). The number of years since the report of ash dieback in the local quadrat was taken as a proxy for the length of the disease presence. It was considered an adequate proxy as local spread of *H. fraxineus* at the scale of a village was shown to be very fast [21]. Leaf necrosis probability was assumed to depend on spring and summer climatic conditions (total rainfall and average temperatures). It is known that H. fraxineus may not survive well during hot summers [29]. In particular, this pathogen is known to stop its growth at temperatures above 28°C and to have limited survival when temperature exceeds 36°C. To account for this, we used TX78, the average daily maximal temperature in July-August (°C). Shoot mortality was assumed to depend on autumn climatic conditions (total rainfalland average temperatures). The meteorological data used were Safran data from Météo-France. Safran data are computed on a 8 x 8 km grid over France [39]. The Safran point closest to each site was selected and daily air temperature and rainfall data were retrieved for the studied years. All predictors were normalised to obtain zero mean and one standard deviation. However, the odds ratio reported in the text are in the non-normalized scale.

The following hierarchical Bayesian model was fitted [40]:

 $logit(pl[i,t]) = \alpha_1 + \alpha_2 x RA6[i,t] + \alpha_3 x TM6[i,t] + \alpha_4 x RA78[i,t] + \alpha_5 x TX78[i,t] + \alpha_6 x LDP[i,t] + Site.l [i]$

logit(**ps**[**i**,**t**]) = $\beta_1 + \beta_2 \mathbf{x} \mathbf{RA912}[\mathbf{i},\mathbf{t}] + \beta_3 \mathbf{x} \mathbf{TM912}[\mathbf{i},\mathbf{t}] + \beta_4 \mathbf{x} \mathbf{TC}[\mathbf{i}] + \beta5 \mathbf{x} \mathbf{pl}[\mathbf{i},\mathbf{t}] + \mathbf{Site.s} [\mathbf{i}]$ with pl[i,t] and ps[i,t] the respective probability of leaf necrosis and shoot mortality in plot i and year t, RA6[**i**,**t**] and TM6[**i**,**t**] the sum of rain and mean daily air temperature in June, RA78[**i**,**t**] and TX78[i,t] the sum of rain and average daily air maximal temperature in July-August (°C), **LDP**[i,t] the number of years of ash dieback presence in the area, RA912[**i**,**t**] and TM912[i,t], the sum of rain and mean daily air temperature in September, October, November and December and TC[i] the percent tree cover within a radius of 100m. Site.l [i] and Site.s [i] are site random factors, taken as a proxy for the site environment (micro climate, site conditions such as host density or infection history).

While ps[i,t] was assumed to follow a binomial distribution, pl[i,t] was assumed to follow a beta distribution as the number of observed leaves per site was very high (from 200 to 400). Non-informative priors were assigned to the model parameters according to a normal distribution N(0, 1e+05) or a uniform distribution U(0,100) for the standard deviation of the random factors. The model was fitted with a burn-in of 10000, for 80000 iterations with a

thinning of 10. Three parallel chains with different initial parameter values were run and convergence was checked by a Gelman-Rubin test (R_{hat} were close to 1).

The suitable periods for spring and autumn were selected in a preliminary analysis. The explored periods targeted either the period of ascospore production in spring or of the development of the shoot necrosis during the autumn. Several periods were tested: from 15 May to 15 June, for the 1 to 30 June or from the 15 June to 15 July for spring and September to October or September to December for autumn. The model showing the best deviance information criteria (DIC) was selected.

The final model was checked by looking at the plot of deviance residuals versus predicted values and by a cross validation procedure. Data were randomly separated in 5 equal groups and the model was fitted using the data from 4 of the groups; predicted leaf necrosis frequency and shoot mortality were computed for the fifth group not used for the fit. The procedure was repeated five times and the observed values obtained by the procedure were plotted against the predicted values.

Relationship between climatic parameters and ability of *H. fraxineus* to produce apothecia in the fructification assay

The proportion of rachises producing apothecia in the fructification assay was analysed by beta regression using the glmmTMB library. The site was included as random factor. Fixed factors were the proportion of leaves with necrosis at the September rating, climatic parameters in the studied sites during the summer (total rainfall and mean of the daily maximal temperature in July and August) and climatic parameters during the overwintering period (average temperature and total rainfall at the overwintering locations from the beginning of October to the end of March).

Supporting information

S1 Table. Studied sites. (DOCX)

S1 Fig. Cross-validation for the model at the plot level relating climatic and site parameters to leak necrosis and shoot mortality probability. A. Leaf necrosis, B. shoot mortality. (DOCX)

S2 Fig. Quantile 0.05 (Q5) and 0.95 (Q95) for the estimated probability of leaf necrosis (A, B) and shoot mortality (C, D). The map were computed using the parameter 0.05 and 0.95 quantiles obtained from the Bayesian procedure fit. The shoot mortality is computed for a forest situation (tree cover of 100%). (DOCX)

Acknowledgments

We wish to thank Olivier Caël and Anaïs Gillet for technical assistance in handling the analysis of rachises ability to produce apothecia and level of infection by *H. fraxineus*. The establishment and rating of the plots in the Charente area was done by Dominique Piou while those of plots from Brittany was done by Laurence Roche from the Département de la Santé des Forêts. We also thank Mireia Gomez-Gallego for her advices in the analysis and for reviewing the manuscript.

Author Contributions

Conceptualization: Benoit Marçais, Claude Husson.

Data curation: Arnaud Giraudel.

Formal analysis: Benoit Marçais, Arnaud Giraudel.

Funding acquisition: Benoit Marçais.

Investigation: Benoit Marçais, Arnaud Giraudel, Claude Husson.

Methodology: Benoit Marçais.

Writing - original draft: Benoit Marçais.

Writing - review & editing: Arnaud Giraudel, Claude Husson.

References

- Cunniffe NJ, Koskella BE, Metcalf CJ, Parnell S, Gottwald TR, Gilligan CA. Thirteen challenges in modelling plant diseases. Epidemics 2015; 10: 6–10. <u>https://doi.org/10.1016/j.epidem.2014.06.002</u> PMID: 25843374
- Sieber TN. Endophytic fungi in forest trees: are they mutualists? Fungal Biology Reviews 2007; 21:75– 89. http://dx.doi.org/10.1016/j.fbr.2007.05.004
- Flowers J, Nuckles E, Hartman J, Vaillancourt L. Latent Infection of Austrian and Scots Pine Tissues by Sphaeropsis sapinea. Plant Disease 2001: 85: 1107–1112. https://doi.org/10.1094/PDIS.2001.85.10. 1107
- Kelnarová I, Černý K, Zahradník D, Koukol O, Sieber T. Widespread latent infection of *Cryptostroma corticale* in asymptomatic *Acer pseudoplatanus* as a risk for urban plantations. For Path 2017; 47: e12344. http://doi.wiley.com/10.1111/efp.12344
- Marsberg A, Kemler M, Jami F, Nagel JH, Postma-Smidt A, Naidoo S et al. *Botryosphaeria dothidea*: a latent pathogen of global importance to woody plant health. Molecular Plant Pathology 2017; 18: 477– 488. https://onlinelibrary.wiley.com/doi/10.1111/mpp.12495
- Pagán I, García-Arenal F. Tolerance of Plants to Pathogens: A Unifying View. Annual Review of Phytopathology 2020; 58: 77–96. https://www.annualreviews.org/doi/10.1146/annurev-phyto-010820-012749 PMID: 32403981
- Alizon S, Michalakis Y. Adaptive virulence evolution: the good old fitness-based approach. Trends Ecol Evol 2015; 30: 248–54. https://doi.org/10.1016/j.tree.2015.02.009 PMID: 25837917
- Davidson JM, Wickland AC, Patterson HA, Falk KR, Rizzo DM. Transmission of *Phytophthora ramorum* in Mixed-Evergreen Forest in California. Phytopathology 2005; 95: 587–596. <u>http://apsjournals.apsnet.org/doi/10.1094/PHYTO-95-0587</u>
- DiLeo MV, Bostock RM, Rizzo DM. *Phytophthora ramorum* Does Not Cause Physiologically Significant Systemic Injury to California Bay Laurel, Its Primary Reservoir Host. Phytopathology 2009; 99:1307– 1311. https://apsjournals.apsnet.org/doi/10.1094/PHYTO-99-11-1307
- 10. Garbelotto M, Schmidt D, Swain S, Hayden K, Lione G. The ecology of infection between a transmissive and a dead-end host provides clues for the treatment of a plant disease. Ecosphere 2017; 8: e01815. http://doi.wiley.com/10.1002/ecs2.1815
- Holdenrieder O, Marco P, Weisberg PJ, Lonsdale D. Tree diseases and landscape processes: the challenge of landscape pathology. Trends in Ecology & Evolution 2004; 19: 446–452. https://doi.org/10. 1016/j.tree.2004.06.003 PMID: 16701303
- 12. Elith J, Leathwick JR. Species Distribution Models: Ecological Explanation and Prediction Across Space and Time. Annual Review of Ecology, Evolution, and Systematics 2009: 40: 677–697. <u>https://www.annualreviews.org/doi/10.1146/annurev.ecolsys.110308.120159</u>
- **13.** Dormann CF, Schymanski SJ, Cabral J, Chuine I, Graham C, Hartig F et al. Correlation and process in species distribution models: bridging a dichotomy: Bridging the correlation-process dichotomy. Journal of Biogeography 2012 39: 2119–2131. https://onlinelibrary.wiley.com/doi/10.1111/j.1365-2699.2011. 02659.x
- 14. Gross A, Holdenrieder O, Pautasso M, Queloz V, Sieber TN. *Hymenoscyphus pseudoalbidus*, the causal agent of European ash dieback. Molecular Plant Pathology 2014; 15: 5–21. https://onlinelibrary.wiley.com/doi/abs/10.1111/mpp.12073

- Cross H, Sønstebø JH, Nagy NE, Timmermann V, Solheim H, Børja I, et al. Fungal diversity and seasonal succession in ash leaves infected by the invasive ascomycete *Hymenoscyphus fraxineus*. New Phytol. 2017 213: 1405–1417. http://doi.wiley.com/10.1111/nph.14204
- Landolt J, Gross A, Holdenrieder O, Pautasso M. Ash dieback due to *Hymenoscyphus fraxineus*: what can be learnt from evolutionary ecology? Plant Path. 2016; 65: 1056–1070. <u>http://doi.wiley.com/10.1111/ppa.12539</u>
- McKinney LV, Nielsen LR, Hansen JK, Kjær ED. Presence of natural genetic resistance in *Fraxinus* excelsior (Oleraceae) to Chalara fraxinea (Ascomycota): an emerging infectious disease. *Heredity* 2011; 106: 788–797. http://www.nature.com/articles/hdy2010119
- McKinney LV, Thomsen IM, Kjaer ED, Nielsen LR. Genetic resistance to *Hymenoscyphus pseudoalbidus* limits fungal growth and symptom occurrence in *Fraxinus excelsior*. For. Path. 2012; 42: 69–74. http://doi.wiley.com/10.1111/j.1439-0329.2011.00725.x
- Schwanda K, Kirisits T. Pathogenicity of Hymenoscyphus fraxineus towards leaves of three European ash species: Fraxinus excelsior, F. angustifolia and F. ornus. Plant Pathology 2016; 65: 1071–1083. http://doi.wiley.com/10.1111/ppa.12499
- 20. Orton ES, Clarke M, Brasier CM, Webber JF, Brown JKM. A versatile method for assessing pathogenicity of *Hymenoscyphus fraxineus* to ash foliage. For. Path. 2019; 49: e12484. https://onlinelibrary.wiley. com/doi/10.1111/efp.12484
- Grosdidier M, Scordia T, Ioos R, Marçais B. Landscape epidemiology of ash dieback. J. Ecol. 2020; 108: 1789–1799. https://onlinelibrary.wiley.com/doi/10.1111/1365-2745.13383
- Grosdidier M, Ioos R., Husson C, Cael O., Scordia T, Marçais B. Tracking the invasion: dispersal of *Hymenoscyphus fraxineus* airborne inoculum at different scales. FEMS Microbiology Ecology, 2018; 94: fiy04. https://academic.oup.com/femsec/article/doi/10.1093/femsec/fiy049/4950393
- Chandelier A, Helson M, Dvorak M, Gischer F. Detection and quantification of airborne inoculum of *Hymenoscyphus pseudoalbidus* using real-time PCR assays. Plant Pathology. 2014; 63: 1296–1305. http://doi.wiley.com/10.1111/ppa.12218
- Semizer-Cuming D, Chybicki IJ, Finkeldey R, Kjær ED. Gene flow and reproductive success in ash (*Fraxinus excelsior* L.) in the face of ash dieback: restoration and conservation. Annals of Forest Science 2021; 78: 14. http://link.springer.com/10.1007/s13595-020-01025-0
- Blodgett JT, Kruger EL, Stanosz GR. Sphaeropsis sapinea and Water Stress in a Red Pine Plantation in Central Wisconsin. Phytopathology, 1997; 87: 429–434. <u>https://apsjournals.apsnet.org/doi/10.1094/</u> PHYTO.1997.87.4.429
- Palmer M, McRoberts R, Nicholls T. Sources of Inoculum of Sphaeropsis sapinea in Forest Tree Nurseries. Phytopathology 1988; 78: 831–835. https://doi.org/10.1094/Phyto-78-831
- Wylder B, Biddle M, King K, Baden R, Webber J. Evidence from mortality dating of *Fraxinus excelsior* indicates ash dieback (*Hymenoscyphus fraxineus*) was active in England in 2004–2005. Forestry 2018; 91: 434–443. https://academic.oup.com/forestry/article/91/4/434/4958744
- Grosdidier M. Épidémiologie de la chalarose du frêne, une maladie causée par l'agent pathogène Hymenoscyphus fraxineus [dissertation]. Nancy: Université de Lorraine (PhD, in french); 2017. <u>https://</u> theses.hal.science/tel-01760779
- **29.** Hauptman T, Piškur B, de Groot M, Ogris N, Ferlan N, Jurk D. Temperature effect on *Chalara fraxinea*: heat treatment of saplings as a possible disease control method. For. Path. 2013; 43: 360–370.
- Grosdidier M, Ioos R, Marçais B. Do higher summer temperatures restrict the dissemination of Hymenoscyphus fraxineus in France? For. Path. 2018; e12426. http://doi.wiley.com/10.1111/efp.12426
- Chumanová E, Romportl D, Havrdová L, Zahradník D, Pešková V, Černý K. Predicting ash dieback severity and environmental suitability for the disease in forest stands. Scandinavian Journal of Forest Research. 2019; 34: 254–266. https://www.tandfonline.com/doi/full/10.1080/02827581.2019.1584638
- Dvorak M, Rotkova G, Botella L. Detection of Airborne Inoculum of *Hymenoscyphus fraxineus* and *H. albidus* during Seasonal Fluctuations Associated with Absence of Apothecia. Forests 2016; 7: 1. <u>http://www.mdpi.com/1999-4907/7/1/1</u>
- Skovsgaard JP, Wilhelm GJ, Thomsen IM, Metzler B, Kirisits T, Havrdová L et al. Silvicultural strategies for Fraxinus excelsior in response to dieback caused by *Hymenoscyphus fraxineus*. Forestry 2017; 90: 455–472. https://academic.oup.com/forestry/article/90/4/455/3749603
- Hietala AM, Timmermann V, Børja I, Solheim H. The invasive ash dieback pathogen *Hymenoscyphus* pseudoalbidus exerts maximal infection pressure prior to the onset of host leaf senescence. Fungal Ecol. 2013; 6: 302–308. https://linkinghub.elsevier.com/retrieve/pii/S1754504813000494
- **35.** Timmermann V, Børja I, Hietala AM, Kirisits T, Solheim H. Ash dieback: pathogen spread and diurnal patterns of ascospore dispersal, with special emphasis on Norway. *EPPO Bull*. 2011; 41: 14–20. http://doi.wiley.com/10.1111/j.1365-2338.2010.02429.x

- 36. Goberville E, Hautekèete N-C, Kirby RR, Piquot Y, Luczak C, Beaugrand G. Climate change and the ash dieback crisis. Scientific Reports 2016; 6: 35303. http://www.nature.com/articles/srep35303 https://doi.org/10.1038/srep35303 PMID: 27739483
- Ioos R, Kowalski T, Husson C, Holdenrieder O. Rapid in planta detection of *Chalara fraxinea* by a realtime PCR assay using a dual-labelled probe. Eur. J. Plant Path. 2009; 125: 329–335. http://link. springer.com/10.1007/s10658-009-9471-x
- **38.** Kowalski T, Holdenrieder O. The teleomorph of *Chalara fraxinea*, the causal agent of ash dieback. For. Path. 2009; 39: 304–308. http://doi.wiley.com/10.1111/j.1439-0329.2008.00589.x
- Quintana Segui P, Le Moigne P, Durand Y, Martin E, Habets F, Baillon M et al. Analysis of near-surface atmospheric variables: Validation of the SAFRAN analysis over France, J. Appl. Meteorol. Climatol. 2008; 47: 92–107. http://journals.ametsoc.org/doi/10.1175/2007JAMC1636.1
- Figueroa-Zúñiga JI, Arellano-Valle RB, Ferrari SLP. Mixed beta regression: A Bayesian perspective. Computational Statistics & Data Analysis 2013; 61: 137–147. <u>http://linkinghub.elsevier.com/retrieve/pii/S0167947312004239</u>
- Marçais B, Kosawang C, Laubray S, Kjaer E, Kirisits T. Ash dieback. In: Asiebu F, Kovalchuk, editors. Forest Microbiology Volume 2: Forest Tree Health. Academic Press, 2022. p. 215–237. ISBN 0323984487, 9780323984485